Author’s response to reviews

Title: The Prevalence of Gastro-Esophageal Reflux Disease and Esophageal Dysmotility in Chinese Patients with Idiopathic Pulmonary Fibrosis

Authors:

Feng Gao (gaofeng4321@sina.com)
Anthony R Hobson (anthonyhobson@hotmail.com)
Zhang M Shang (shangmed@163.com)
Yan X Pei (peiyanxiang1972@126.om)
Yan Gao (gaoxiaocong07@sina.com)
Jian X Wang (1wjx@163.com)
Wan N Huang (lg429@sina.com)

Version: 2 Date: 13 March 2014

Author’s response to reviews:

My manuscript includes line and page numbers, and the authors’ email this time.